Author's response to reviews

Title: Unanswered clinical questions in the management of cardiometabolic risk in the elderly: a statement of the Spanish Society of Internal Medicine

Authors:

Ricardo Gomez-Huelgas (ricardogomezhuelgas@hotmail.com)
Vicente Giner-Galvan (ginervicgal@gmail.com)
Jose M Mostaza (josemaria.mostaza@salud.madrid.org)
Jose I Cuende (jcuendem@medynet.com)
Jose M de Miguel-Yanes (josemaria.demiguel@salud.madrid.org)
Eduardo Rovira (erovira@hospital-ribera.com)
Demetrio Sanchez-Fuentes (sanchezfuentes@msn.com)
Carmen Suarez Fernandez (csuarez.hlpr@salud.madrid.org)
Pilar Roman Sanchez (roman_pil@gva.es)

Version: 3 Date: 8 December 2014

Author's response to reviews:

Dear Mr. Gilbert Tacbobo,

Answering to your request for additional modification for the abovementioned manuscript, please find below the point-by-point description of the changes made:

1) Title Page: we have added the email address of each author. With these data, the <author names/author emails addresses> in the manuscript are consistent with those entered in the submission system.

2) Abstract: we have confirmed that the abstract entered in the submission system and the abstract in the manuscript file are identical. We wonder if you mean that this abstract is different than the one submitted initially (October 4th 2014), which is true, since we modified the content (including a “methods” section) and the English style, as it was suggested by the reviewers. In any case, since the structure of the “Debate” section only include Background, Discussion and Summary, we have modified it accordingly.

3) We have removed all authors’ suffixes from title page.

4) We have also specified more in detail the Authors’ Contributions in the corresponding section.

We sincerely hope that the current version of the manuscript will be acceptable for publication in your journal.

Sincerely

Prof. Ricardo Gómez-Huelgas